

# 

**Citation:** França CT, Hostetler JB, Sharma S, White MT, Lin E, Kiniboro B, et al. (2016) An Antibody Screen of a *Plasmodium vivax* Antigen Library Identifies Novel Merozoite Proteins Associated with Clinical Protection. PLoS Negl Trop Dis 10(5): e0004639. doi:10.1371/journal.pntd.0004639

**Editor:** Christian R. Engwerda, Queensland Institute of Medical Research, AUSTRALIA

Received: December 28, 2015

Accepted: March 29, 2016

Published: May 16, 2016

**Copyright:** This is an open access article, free of all copyright, and may be freely reproduced, distributed, transmitted, modified, built upon, or otherwise used by anyone for any lawful purpose. The work is made available under the <u>Creative Commons CC0</u> public domain dedication.

Data Availability Statement: Data has been deposited in Dryad: dryad.pn8mn">dryad. pn8mn">dryad.pn8mn">http://dx.doi.org/10.5061/\_ tmp\_url\_2\_8mn. PNG data cannot be made publicly available because it would compromise participant privacy and violates the ethical agreement in the informed consent forms. Data is available upon reasonable request by contacting the PNG Medical Research Advisory Committee and the PNG Institute of Medical Research IRB. The contact is Dr. William Pomat, secretary PNG IMR IRB: William.Pomat@pngimr.org.pg. RESEARCH ARTICLE

# An Antibody Screen of a *Plasmodium vivax* Antigen Library Identifies Novel Merozoite Proteins Associated with Clinical Protection

Camila T. França<sup>1,2</sup>, Jessica B. Hostetler<sup>3,4</sup>, Sumana Sharma<sup>4</sup>, Michael T. White<sup>1,5</sup>, Enmoore Lin<sup>6</sup>, Benson Kiniboro<sup>6</sup>, Andreea Waltmann<sup>1,2</sup>, Andrew W. Darcy<sup>7</sup>, Connie S. N. Li Wai Suen<sup>1,2</sup>, Peter Siba<sup>6</sup>, Christopher L. King<sup>8</sup>, Julian C. Rayner<sup>4‡\*</sup>, Rick M. Fairhurst<sup>3‡\*</sup>, Ivo Mueller<sup>1,2,9‡\*</sup>

1 Population Health and Immunity Division, Walter and Eliza Hall Institute of Medical Research, Melbourne, Australia, 2 Department of Medical Biology, University of Melbourne, Melbourne, Australia, 3 Laboratory of Malaria and Vector Research, National Institute of Allergy and Infectious Diseases, National Institutes of Health, Bethesda, Maryland, United States of America, 4 Malaria Programme, Wellcome Trust Sanger Institute, Hinxton, United Kingdom, 5 Center for Outbreak Analysis and Modelling, Department of Infectious Disease Epidemiology, Imperial College London, London, United Kingdom, 6 Vector Borne Diseases Unit, PNG Institute of Medical Research, Madang, Papua New Guinea, 7 National Health Training & Research Institute, Ministry of Health, Honiara, Solomon Islands, 8 Center for Global Health & Diseases, Case Western Reserve University, Cleveland, Ohio, United States of America, 9 ISGlobal, Barcelona Ctr. Int. Health Res. (CRESIB), Hospital Clínic, Universitat de Barcelona, Barcelona, Spain

‡ These authors are joint senior authors on this work.
\* jr9@sanger.ac.uk (JCR); rfairhurst@niaid.nih.gov (RMF); ivomueller@fastmail.fm (IM)

# Abstract

# Background

Elimination of *Plasmodium vivax* malaria would be greatly facilitated by the development of an effective vaccine. A comprehensive and systematic characterization of antibodies to *P. vivax* antigens in exposed populations is useful in guiding rational vaccine design.

# Methodology/Principal Findings

In this study, we investigated antibodies to a large library of *P. vivax* entire ectodomain merozoite proteins in 2 Asia-Pacific populations, analysing the relationship of antibody levels with markers of current and cumulative malaria exposure, and socioeconomic and clinical indicators. 29 antigenic targets of natural immunity were identified. Of these, 12 highly-immunogenic proteins were strongly associated with age and thus cumulative lifetime exposure in Solomon Islanders (*P*<0.001–0.027). A subset of 6 proteins, selected on the basis of immunogenicity and expression levels, were used to examine antibody levels in plasma samples from a population of young Papua New Guinean children with well-characterized individual differences in exposure. This analysis identified a strong association between reduced risk of clinical disease and antibody levels to P12, P41, and a novel hypothetical protein that has not previously been studied, PVX\_081550 (IRR 0.46–0.74; P<0.001-0.041).

Funding: This study was funded in part by the Southwest Pacific International Centre of Excellence in Malaria Research (NIH grant U19Al089686 "Research to control and eliminate malaria in the Southwest Pacific"), the National Institutes of Health (AI063135), the Intramural Research Program, National Institute of Allergy and Infectious Diseases, National Institutes of Health), the National Health & Medical Research Council (#1021544), the Malaria Elimination Science Alliance (MESA), the Wellcome Trust (#098051), and the UK Medical Research Council (MR/J002283/1). This work was also made possible through Victorian State Government Operational Infrastructure Support and Australian Government NHMRC IRIISS. IM is supported by an NHMRC Senior Research Fellowship (#1043345) and CTF is supported by the University of Melbourne - Melbourne International Postgraduate Scholarship. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing Interests:** The authors have declared that no competing interests exist.

# **Conclusion/Significance**

These data emphasize the benefits of an unbiased screening approach in identifying novel vaccine candidate antigens. Functional studies are now required to establish whether PVX\_081550 is a key component of the naturally-acquired protective immune response, a biomarker of immune status, or both.

# Author Summary

*Plasmodium vivax* is now the predominant malaria parasite outside Africa. Because P. vivax can remain dormant in the liver for months, identifying and treating P. vivax in asymptomatic individuals is difficult. Additionally, current widely-used vector control measures are less efficient against mosquitoes that transmit *P. vivax*. An effective vaccine would therefore immensely facilitate P. vivax elimination. Unfortunately, little is known about P. vivax biology and only a few proteins have been investigated as targets for vaccine development. To address these knowledge gaps, we measured antibody levels to 34 entire ectodomain proteins predicted to be involved in *P. vivax* invasion of erythrocytes, in samples from individuals living in 2 malaria-endemic Asia-Pacific countries. We found that antibodies in malaria-exposed Solomon Islanders were reactive to the majority of proteins in our panel, and that antibodies to 12 of these proteins strongly reflected cumulative lifetime exposure to P. vivax. In samples from Papua New Guinea children, we identified an association between antibodies to 3 proteins and protection against clinical malaria. Our results demonstrate that screening antibodies to a large number of P. vivax proteins is a useful approach in identifying novel targets of immunity. Functional studies are now required to establish whether these proteins are biomarkers of an individual's immune status, potential vaccine candidates that warrant further development, or both.

# Introduction

Intensified research and funding have helped to significantly reduce the morbidity and mortality of malaria, and an increasing number of countries are now aiming to eliminate this disease [1-3]. In Asia-Pacific and the Americas, however, interrupting local *Plasmodium vivax* transmission will be particularly challenging. The ability of *P. vivax* to form dormant liver hypnozoites, which are responsible for ~80% of all blood-stage infections [4, 5], provides a source of new blood-stage infections in the absence of transmission. *P. vivax* commonly causes low-density asymptomatic infections that often go undetected and thus untreated. Moreover, the early maturation and peripheral circulation of *P. vivax* gametocytes, coupled with high infectivity and rapid development in mosquitoes, make *P. vivax* more refractory to control measures [6]. As a consequence, *P. vivax* is now the predominant *Plasmodium* species outside Africa [1].

New tools are needed to control and eliminate *vivax* malaria. Vector control strategies that are broadly effective in reducing *P. falciparum* transmission, such as insecticide-treated nets (ITNs) and indoor residual spraying, seem to be less effective against *P. vivax* vectors [7, 8], which are more likely to bite and rest outdoors, and less likely to bite humans than African *P. falciparum* vectors [9]. Furthermore, primaquine, the only drug effective against *P. vivax* hypnozoites, is associated with hemolysis in glucose-6-phosphate dehydrogenase-deficient individuals [10]. Similar effects have been seen for tafenoquine, the only other liver-stage drug in clinical development [11]. Given these challenges, the development of a highly effective vaccine

would immensely facilitate *P. vivax* elimination, perhaps even more so than *P. falciparum* elimination [12].

Merozoites represent the only extracellular phase of the *Plasmodium* blood-stage life cycle, and merozoite antigens are therefore appropriate vaccine targets. Several studies have investigated merozoite antigens as targets of natural protective immunity to *P. falciparum* malaria [13], and their potential as vaccine candidates [14]. For *P. vivax*, the availability of the genome sequence [15] and transcriptome [16] have enhanced our understanding of this parasite's biology, facilitating the identification of many proteins that are homologous to *P. falciparum* antigens [17–19]. However, the targets of natural immunity to *P. vivax* malaria remain poorly understood, and systematic screens of multiple antigens are lacking [20]. As a consequence, there are currently only a handful of *P. vivax* vaccine candidate antigens in pre-clinical development, with only a single blood-stage antigen (PvDBP) nearing clinical development [21].

In this study, we investigated 34 recombinant *P. vivax* protein ectodomains [22], known or predicted to localize to the merozoite cell surface, micronemes, or rhoptries, as targets of natural immunity. For 12 highly-immunogenic proteins, we investigated associations between levels of antibodies and indicators of current and cumulative malaria exposure in a moderately-endemic area of the Solomon Islands (SI). Using a cohort of young Papua New Guinean (PNG) children with well-characterized individual differences in exposure, we identified an association between reduced incidence of clinical disease and antibody levels to 3 proteins, including a novel hypothetical protein that has not been previously studied. These data emphasize the benefits of an unbiased screening approach in identifying vaccine candidates and indicate that these 3 antigens are high-priority targets for further functional studies, and potentially vaccine development.

#### Methods

#### **Protein library**

Proteins were designed, constructed, and expressed as described previously for *P. falciparum* merozoite proteins [23, 24]; the *P. vivax* ectodomain library has been described in detail by Hostetler et al. (S1 Table) [22]. Briefly, sequences derived from the *P. vivax* Salvador-1 strain encoding merozoite ectodomains, excluding their signal peptide, transmembrane domain, and glycosylphosphatidylinositol (GPI) anchor sequences (if present), were codon-optimized for expression in human cells and chemically synthesized (GeneArt AG). Soluble recombinant proteins (S1 Table) containing a ~25-kDa C-terminal rat Cd4d3+d4 (Cd4) tag were expressed in human embryonic kidney (HEK) 293E cells as either biotinylated or 6-His-tagged forms, culture supernatants were collected 6 days after transfection, and biotinylated proteins were dialysed in HEPES-buffered saline. All expression plasmids are openly available at Addgene (http://www.addgene.org/express/vivax/).

#### Protein purification

6-His-tagged proteins were purified by immobilized metal-ion affinity chromatography using HisTrap-HP columns on an AKTA Xpress (GE Healthcare) following the manufacturer's instructions. Proteins were then conjugated to Luminex Microplex microspheres (Luminex Corporation) as described [25], using the following concentrations per  $2.5 \times 10^6$  beads: P41,  $0.5 \mu$ g/mL; PVX\_081550,  $1.2 \mu$ g/mL; P12,  $0.2 \mu$ g/mL; GAMA,  $0.015 \mu$ g/mL; ARP,  $0.09 \mu$ g/mL; CyRPA,  $1.5 \mu$ g/mL; and Cd4,  $2 \mu$ g/mL. Coupling efficiency was determined by using an immune plasma pool known to be highly reactive with the antigens, with the appropriate antigen concentration resulting in high fluorescence intensity by the reporter fluorochrome.

#### Study populations

Immunoreactivity and comprehensive screens. Samples collected in a cross-sectional survey (3501 individuals aged  $\geq 6$  months) in May 2012 in Ngella, Central Island Province, SI were used [26]. A random subset of 22 adolescents (10–19 years) and 24 adults (20–50 years) were selected for the immunoreactivity screen, because they had a higher cumulative exposure and were thus more likely to have acquired substantial levels of natural immunity. In all 46 samples, total IgG to 34 biotinylated proteins and Cd4-tag alone bound to streptavidin-coated plates was measured using ELISA, as described [22].

12 highly-immunoreactive proteins identified in the immunoreactivity screen were subsequently screened using ELISA [22] in 144 individual samples from the same survey to investigate relationships with infection status, clinical symptoms, and socioeconomic indicators. Samples were randomly selected based on age and infection status in a 3x3 factorial design that included 48 children (5–9 years), 48 adolescents (10–19 years), and 48 adults (20–80 years) either without any *Plasmodium* infections, with a current *P. vivax* monoinfection detected by PCR, or with a current *P. vivax* monoinfection detected by both PCR and light microscopy (LM). A detailed description can be found in <u>S2 Table</u>. Plasma pools from malaria-naïve Australian and highly-immune PNG adult donors were included on each plate as negative and positive controls, respectively. Samples were tested in duplicate on separate plates.

**Cohort study**. Of the 12 highly-immunoreactive proteins, 6 were selected based on their reactivity in Solomon Islanders and previously reported reactivity in Cambodians [22], as well as their expression levels. These 6 proteins were expressed and purified as described above, along with a Cd4-tag control, and used to measure total IgG in samples from a longitudinal cohort of PNG children described in detail in [27]. Briefly, 264 children aged 1-3 years from a rural area near Maprik, East Sepik Province were enrolled in March-September 2006 and followed for up to 16 months. Children were actively checked for morbidity every 2 weeks, and passive case detection was performed over the entire study period. All PCR+ P. vivax infections were genotyped to determine the incidence of genetically distinct blood-stage infections acquired during follow-up (i.e., the molecular force of blood-stage infections, molFOB) [28, 29]. Only samples from 230 children who completed follow-up were included in the present study. Luminex bead array assays to measure total IgG were performed as described [30] using plasma and secondary antibody donkey F(ab')<sub>2</sub> anti-human IgG Fc R-PE (Jackson ImmunoResearch) diluted 1:100 in PBS. Bead array assays included the same set of controls used for ELI-SAs and a dilution series of the highly-immune PNG positive control pool to standardize plateto-plate variations. Samples were tested in singlicate.

#### Statistical analysis

**Immunoreactivity screen.** For all ELISA data, an OD cut-off of 0.1 was set as a conservative lower limit based on the plate reader's limit of accurate detection, and samples with OD values <0.1 were set to 0.1. Differences in population mean antibody levels to proteins and Cd4 alone, and by age groups were assessed using Mann-Whitney U test.

**Comprehensive screen**. Duplicate wells were averaged and OD values for Cd4 subtracted to correct for background. OD values were log<sub>10</sub>-transformed and differences in mean antibody levels by exposure variables assessed using 2-tailed unpaired *t*-test or ANOVA. Multivariate ANOVA models were fitted including all variables that were univariately associated with IgG levels, with the best model determined by backward elimination using Wald's Chi-square tests for individual variables.

Cut-offs for positivity were set at 2 standard deviations above the mean antibody levels to the negative controls. Differences in the breadth of antibody levels by age and infection status

were assessed using negative binomial regression. To estimate seroconversion and seroreversion rates, seroprevalence data were stratified in 5- or 10-year age bins and analysed using reverse catalytic modelling as described elsewhere [<u>31</u>, <u>32</u>], with the model fitted in a Bayesian framework.

**Cohort study**. To correct plate-to-plate variations, the dilutions of the highly-immune PNG positive control pool were fitted as plate-specific standard curves using a 5-parameter logistic regression model [33]. For each plate, Luminex median fluorescence intensity (MFI) values were interpolated into relative antibody units based on the parameters estimated from the plate's standard curve. Antibody units ranged from 1.95x10<sup>-5</sup> (i.e., equivalent to 1:51200 dilution of the immune pool) to 0.02 (1:50). To account for the background reactivity to the Cd4-tag, antibody levels were re-scaled by using linear regression to estimate the antibody levels that would be detected if reactivity to the Cd4-tag was zero, as follows:

$$log(AB_meas) = log(AB_true) + \beta^* log(Cd4)$$

where AB\_meas = measured antibody level, AB\_true = true antibody level to a given antigen, and Cd4 = measured antibody level to the Cd4-tag.

Associations between antibodies and age and exposure were assessed using Spearman rank correlation, and differences with infection using 2-tailed unpaired t-test on log<sub>10</sub>-transformed values. Negative binomial GEE models with exchangeable correlation structure and semirobust variance estimator [34] were used to analyse the relationship between IgG levels and prospective risk of *P. vivax* episodes (defined as axillary temperature  $\geq$  37.5°C or history of fever in the preceding 48 hours with a current P. vivax parasitemia >500 parasites/µL). For this, IgG levels were classified into terciles and analyses done by comparing children with low versus medium and high antibody levels. Children were considered at risk from the first day after the initial blood sample was taken. The molFOB, representing individual differences in exposure, was calculated as the number of new P. vivax clones acquired per year at risk, and square-root transformed for better fit [29]. All GEE models were adjusted for seasonal trends, village of residency, age, and molFOB. In multivariate models that included all antigens that were univariately associated with protection, the best model was determined by backward elimination using Wald's Chi-square tests for individual variables. To investigate the effect of increasing cumulative IgG levels to the combination of antigens on the risk of *P. vivax* episodes, we assigned a score of 0, 1, and 2 to low, medium, and high antibody levels, respectively, and then added up the scores to the 6 antigens to generate a breadth score per child. The breadth score was then fitted as a continuous covariate in the GEE model described above. All analyses were performed using STATA version 12 (StataCorp) or R version 3.2.1 (http://cran.r-project. org).

#### Ethical statement

Ethical clearance was obtained from the PNG Medical Research and Advisory Committee of the Ministry of Health, Solomon Islands National Health Research Ethics Committee, and the Walter and Eliza Hall Institute. Informed consent was obtained from all participants and in cases of children from their parents or guardians. As approved by the Australian and Solomon Islands' ethics committees, only verbal consent, documented on each participant's case report form, was obtained from participants in the cross-sectional survey in Ngella, Solomon Islands. Written informed consent was obtained from the parents or guardians of all children participating in the PNG cohort study.

#### Results

# P. vivax merozoite proteins are targets of natural humoral immunity

We first investigated whether IgG from 46 *P. vivax*-exposed SI individuals recognized antigens from our library. There was a high degree of variability in IgG levels to the different proteins. The population mean antibody level to 85.3% (29/34) of proteins was significantly higher than to Cd4 alone (P<0.001–0.018) and these were therefore considered immunogenic (Fig 1). IgG levels were similar between adolescents and adults, except for 7/34 proteins (P<0.001–0.032) (Fig 1). We then selected 12 of the most immunogenic proteins, MSP3.3, MSP10, MSP7.6, MSP3.10, P12, ARP, P41, MSP5, GAMA, RIPR, MSP1, and CyRPA, for further analysis.

# The breadth of antibodies increases with age and infection

We tested these 12 proteins against a larger panel of 144 SI samples. Antibody seroprevalence ranged from 31.3% (P41) to 100% (MSP1) (<u>S1 Table</u>). MSP1 was recognized by 100% of samples, and P12, GAMA, MSP3.10, and RIPR were each recognized by at least 85% of the children, 89% of the adolescents, and 93% of the adults (<u>Fig 2A</u>). Children recognized fewer proteins (mean 7.06) than adults (mean 9.65; P<0.001) or adolescents (mean 8.08; P = 0.051) (<u>Fig 2C</u>). Similarly, individuals with a current infection detected by both PCR and LM (mean 8.90; P = 0.005), but not those with only PCR+ infections (mean 8.33; P = 0.140), had antibodies to significantly more



**Fig 1. IgG reactivity to 34** *P. vivax* merozoite proteins in Solomon Islanders. Boxplots show median optical density (horizontal bar), interquartile range (boxes), range (whiskers), and outliers (open circles). Antibodies to the Cd4-tag are represented as mean (solid red line) and mean + 2 standard deviations (dashed red line). Clear boxes represent adolescents (10–19 years; n = 22;) and striped boxes represent adults (20–50 years; n = 24). Asterisks indicate 29/34 proteins for which the population mean antibody level was significantly higher than to Cd4 alone. ¶ symbols indicate 7/34 proteins for which IgG levels differed significantly between adolescents and adults. *P* values are from Mann-Whitney U tests and were deemed significant if <0.05.

doi:10.1371/journal.pntd.0004639.g001



**Fig 2.** Seroprevalence profiles for 12 *P. vivax* merozoite antigens in Solomon Islanders. For each antigen, positivity was defined as 2 standard deviations above the mean optical density from Australian malaria-naïve adults. (A) and (B) show heatmaps of seroprevalence by age group (5–9 years, n = 48; 10–19 years, n = 48; 20–80 years, n = 48) and by *P. vivax* infection status (Not infected, n = 48; PCR+, n = 48; PCR+ LM+ n = 48), respectively. Each row shows antibodies observed in all individuals to a single protein, and each column shows antibodies observed in a single individual to all proteins. (C) and (D) show density plots representing the number of antigens recognized in each of the age and infection status groups, respectively. Normally-distributed density curves are shown in dark green. *P* values are from negative binomial regression and were deemed significant if <0.05.

doi:10.1371/journal.pntd.0004639.g002

proteins than noninfected individuals (mean 7.56), suggesting a limited effect of recent infections even in adults (Fig 2B and 2D). We applied a serocatalytic model to the seroprevalence data to investigate the kinetic (seroconversion and seroreversion rates) of IgG to antigens with <85% prevalence in children. The estimates are shown in S1 Text.

#### Antibody levels reflect both cumulative and current exposure

The magnitude of the cumulative levels (i.e., sum of IgG levels to all antigens, per individual), as well as IgG levels to all individual antigens were strongly associated with age, increasing



**Fig 3. Magnitude of IgG levels to 12** *P. vivax* merozoite proteins in Solomon Islanders. Boxplots show median optical density (horizontal bar), interquartile range (boxes), range (whiskers), and outliers (open circles). (A) shows data according to age group (5–9 years, n = 48, clear boxes; 10–19 years, n = 48, grey boxes; 20–80 years, n = 48, striped boxes). (B) shows data according to *P. vivax* infection status (Not infected, n = 48, clear boxes; PCR+, n = 48, grey boxes; PCR+, n = 48, striped boxes). *P* values are from ANOVA and were deemed significant if <0.05.

doi:10.1371/journal.pntd.0004639.g003

significantly from children to adults (P < 0.001-0.027) (Fig 3A). The magnitude of the cumulative antibody levels was also increased in the presence of current infection at either lower (PCR + only, P = 0.008) or higher parasite density (PCR+ LM+, P = 0.001) (Fig 3B). Individually, lower-density infections were associated with higher IgG levels to CyRPA (P = 0.022), GAMA (P = 0.015), and MSP3.10 (P = 0.013) only, while higher-density infections were associated with higher IgG levels to a larger number of antigens: CyRPA (P < 0.001), GAMA (P = 0.015), MSP3.10 (P = 0.020), P41 (P = 0.035), MSP1 (P = 0.020), MSP10 (P = 0.039), and RIPR (p = 0.019) (Fig 3B). These data indicate that the antigens in our panel are good markers of cumulative exposure, and that some of them are also markers of current infection.

#### Predictors of antibody levels to P. vivax merozoite proteins

In multivariate analysis, age remained strongly associated with increased cumulative and IgG levels to all individual antigens (P<0.001–0.023) (<u>S2 Table</u>). Lower-density infections remained associated with increased IgG levels to CyRPA (P = 0.012) and GAMA (P = 0.007), and higher-density infections remained associated with increased IgG levels to CyRPA (P<0.001), GAMA

(P = 0.005), P12 (P = 0.017), P41 (P = 0.031), MSP1 (P = 0.014), and RIPR (P = 0.012). For MSP5, IgG levels were higher only in adults with a current infection (P = 0.022) (S2 Table).

We compared IgG levels against a large number of other epidemiological variables (e.g., region, clinical symptoms, and socioeconomic indicators), but none of them were significantly associated with differences in antibody levels for any antigen (S2 Table). The use of ITNs was the only variable that had any significant association, with the use of ITNs in previous years associated with reduced IgG levels to RIPR (P = 0.030), MSP1 (P = 0.024), and MSP3.3 (P = 0.015) (S2 Table). If ITN use is considered a marker for exposure, this also indicates that the levels of antibodies targeting these antigens are particularly sensitive to recent exposure.

# Associations of antibody levels with cumulative and current exposure in PNG children

To establish whether these associations were population-specific or more broadly generalizable, we tested IgG levels to a subset of 6 antigens, chosen on the basis of immunoreactivity and expression levels, in a sub-cohort of 230 PNG children. The median age of the population was 1.7 years (IQR 1.3–2.5), and the prevalence of *P. vivax* infection at baseline was 55% by PCR. IgG levels to ARP, CyRPA, and PVX\_081550 were positively associated with age (r = 0.15– 0.25; *P* = 0.001–0.027). For PVX\_081550, stronger increases in IgG with age were observed in children without current infections (r = 0.33; *P*<0.001) than with current infections (r = 0.18; *P* = 0.048) (<u>S3 Table</u>). A current *P. vivax* infection was associated with higher IgG levels to CyRPA (*P*<0.001), P12 (*P*<0.001), P41 (*P* = 0.001), and PVX\_081550 (*P* = 0.001) (<u>S3 Table</u>). When considering cumulative exposure as a product of age and the number of *P. vivax* infections acquired over time (molFOB), increasing IgG levels with cumulative exposure to PVX\_081550 (r = 0.41 *P*<0.001) and CyRPA (r = 0.14, *P* = 0.032) are observed in children without current infections (<u>S3 Table</u>).

# Antibody levels and risk of P. vivax malaria

During the 16 months of follow-up of the PNG cohort, children experienced an IRR of 1.25 ( $_{95\%}$ CI 1.08–1.45) malaria episodes with *P. vivax* >500 parasites/µL/year at risk. We applied the unadjusted GEE model to test whether responses to specific antigens were associated with a reduced risk of infection. Children with high levels of IgG to PVX\_081550 (IRR<sub>H</sub> 0.41; *P*<0.001) and P41 (IRR<sub>H</sub> 0.63; *P* = 0.019) both had a significantly lower risk of clinical *P. vivax* malaria (Table 1). When adjusting for confounders, medium and high levels of IgG to PVX\_081550 (IRR<sub>M</sub> 0.74, *P* = 0.041; IRR<sub>H</sub> 0.46, *P*<0.001), and high IgG levels to P41 (IRR<sub>H</sub> 0.56; *P*<0.001) and P12 (IRR<sub>H</sub> 0.65; *P* = 0.012) were associated with protection. No association with protection was observed for levels of IgG to GAMA, CyRPA, and ARP (Fig 4; Table 1).

IgG levels to the 3 antigens associated with protection were significantly correlated (r = 0.34–0.66; *P*<0.001) (<u>S4 Table</u>), suggesting co-acquisition. In multivariate analyses, only high levels of IgG to PVX\_081550 remained strongly associated with reduced risk of *P. vivax* episodes (IRR<sub>H</sub> 0.54; *P* = 0.001), indicating that this antigen may be a key target of natural immunity or a good marker of immunity. There were no significant associations between levels of IgG to any of these 3 antigens and risk of clinical episodes caused by *P. falciparum* with any parasite density (IRR 0.92–1.18; *P*>0.10) (<u>S6 Table</u>).

#### Protection increases with increasing antibody repertoire

There was a very strong association between increasing antibody repertoire and increase in protection. Each increase in 1 unit of the breadth score (described in Methods) was associated with a reduction of approximately 7% in the risk of *P. vivax* episodes (IRR 0.93; 95%CI 0.90–0.97;

Antigen	ulRR	95%CI		P value	alRR*	95%CI		P value
PVX_081550 M	0.76	0.54	1.05	0.10	0.74	0.55	0.99	0.041
PVX_081550 H	0.41	0.29	0.60	<0.001	0.46	0.33	0.64	<0.001
ARP M	0.93	0.66	1.32	0.68	0.98	0.73	1.32	0.91
ARP H	1.00	0.69	1.46	0.98	0.88	0.63	1.23	0.47
GAMA M	1.12	0.80	1.57	0.51	1.03	0.75	1.40	0.87
GAMA H	0.82	0.55	1.23	0.34	0.75	0.54	1.04	0.08
P41 M	0.96	0.68	1.36	0.83	0.89	0.67	1.18	0.41
P41 H	0.63	0.43	0.93	0.019	0.56	0.41	0.77	<0.001
P12 M	1.05	0.75	1.47	0.79	0.96	0.71	1.29	0.77
P12 H	0.69	0.47	1.02	0.06	0.65	0.47	0.91	0.012
CyRPA M	1.06	0.76	1.46	0.74	1.17	0.88	1.57	0.28
CyRPA H	0.88	0.59	1.30	0.52	0.81	0.58	1.12	0.20

Table 1. Association between levels of IgG to *P. vivax* merozoite proteins and protection against clinical malaria (with parasite density >500/µL of blood) in Papua New Guinean children.

Abbreviations: M = Medium antibody levels; H = High antibody levels; 95%CI = 95% confidence interval; uIRR = Unadjusted incidence rate ratio; aIRR = Adjusted incidence rate ratio.

\*Adjusted for individual differences in exposure (molFOB), age, village of residence, and season.

P values, uIRR, and aIRR from negative binomial GEE models. P values <0.05 were deemed significant.

doi:10.1371/journal.pntd.0004639.t001

P = 0.001). However, once we accounted for differences in IgG to PVX\_081550, the breadth effect was no longer significant (IRR 0.98; 95%CI 0.93–1.04; P = 0.49), while high levels of IgG to PVX\_081550 remained associated with protection (IRR<sub>H</sub> 0.51; 95%CI 0.33–0.80; P = 0.004). This finding suggests that IgG level to PVX\_081550 is a key marker of protective immunity.

#### Discussion

The discovery and rational prioritization of *P. vivax* proteins as candidates for inclusion in a future *P. vivax* vaccine would be greatly facilitated by a comprehensive and systematic characterization of antibody response to *P. vivax* antigens in exposed individuals. Although



Fig 4. IgG to 6 *P. vivax* merozoite proteins and risk of clinical malaria in PNG children. Data are plotted as incidence rate ratios and 95% confidence intervals, adjusted for exposure (molFOB), age, season, and village of residency. Clinical malaria was defined as axillary temperature  $\geq$ 37.5°C or history of fever in the preceding 48 hours with a current *P. vivax* parasitemia >500 parasites/µL. *P* values are from negative binomial GEE models and were deemed significant if <0.05.

doi:10.1371/journal.pntd.0004639.g004

epidemiological associations do not necessarily denote causality, the identification of such 'protective' antibody targets in naturally exposed individuals can be used to prioritize antigens or antigen combinations before testing their efficacy and thus potential vaccine suitability in functional studies. Such sero-epidemiological discovery and down-selection are particularly important for *P. vivax*, where the lack of stable *in-vitro* culture and genetic manipulation techniques [6] make functional studies and biology-drive discovery difficult, low throughput, and thus very expensive. To date, only a very small number of *P. vivax* antigens, such as DBP, MSP1, MSP3, MSP9, and AMA1 have been investigated [21, 35, 36]. The complexity of naturallyacquired immunity against P. vivax [37] and the likelihood that it's multifactorial and involves antibodies against several antigenic targets, unlikely to be identified in only one study, highlight the importance of conducting more screening studies. Investigating the large number of potential targets found in the parasite proteome, however, has been constrained in large part by the difficulty of producing natively-folded recombinant P. vivax antigens. We have leveraged our recent development of a large library of immunoreactive merozoite surface and secreted entire ectodomain proteins [22] to perform systematic studies of reactivity to P. vivax blood-stage antigens in 2 Asia-Pacific populations.

The vast majority of these proteins (28/34) were recognized by plasma IgG from asymptomatic (including noninfected) adolescent and adult Solomon Islanders. Of these, 27 were also recognized by pooled IgG from Cambodian *P. vivax* malaria patients [22]. Although there are individual differences between study populations (e.g., PVX\_116675 was only recognized in SI, and PVX\_110950 and RhopH3 were only recognized in Cambodia), the use of a large protein library for the first time confirms the broad immunogenicity of a large number recombinant proteins, and also that the pool of potential vaccine targets is much deeper than has been studied to date.

For 12 highly-immunogenic proteins (MSP3.3, MSP10, MSP7.6, MSP3.10, P12, ARP, P41, MSP5, GAMA, RIPR, MSP1, and CyRPA), we confirmed that IgG levels increase more strongly with age, and thus cumulative life-time exposure, than with current infection. Lower and asymptomatic parasitemias are prevalent in SI, a sign that despite significant recent reductions in transmission, residents have acquired significant immunity that is characterized by long-lasting, stable antibody levels. In several studies, antibodies to the *P. falciparum* homologs of some of the proteins included in our study were shown to be strongly associated with clinical immunity to *P. falciparum* [<u>38</u>, <u>39</u>]. It is therefore likely that the observed high antibody levels to these *P. vivax* proteins contribute to the strong levels of clinical immunity in the SI community.

The associations of clinical immunity with antibodies to 3 antigens (P12, P41, and PVX\_081550) were confirmed in a cohort of young, semi-immune PNG children. The observed reductions in risk of *P. vivax* malaria were comparable to those associated with high antibody titers to *P. vivax* MSP3 $\alpha$  and MSP9 [34]. In *P. falciparum*, P12 is a GPI-anchored rhoptry protein [40], while P41 is localized to the merozoite surface [41]; together, they form a heterodimer and are thought to be involved in reticulocyte invasion, although neither is essential for parasite growth *in vitro* [42]. Both are strongly recognized by natural immunity, and antibodies have also been associated with clinical protection [38, 39]. It is likely that *P. vivax* P12 and P41, which also form a heterodimer [22], have comparable functions.

The protein with the strongest association with protection was the hypothetical protein, PVX\_081550. Its *P. falciparum* homologue has recently been identified as StAR-related lipid transfer protein [43], able to transfer different lipids between phospholipid vesicles. In *P. falciparum*, this protein localizes to the parasitophorous vacuole (PV); there is some evidence that it may be transferred into the apical organelles of mature merozoites, where it may play a role in forming the PV during the invasion process [43]. Although the *P. falciparum* protein was also

found to be immunogenic [44], it is unclear whether antibodies to it interfere with parasite function (e.g., block erythrocyte invasion) or are simply elicited by proteins released from the PV upon schizont rupture and thus serve only as markers of an individual's immune status. Both proteins are polymorphic, with nonsynonymous/synonymous SNP ratios of 1.9–2.3 (PlasmodDBv26 [45]). Further studies are now needed to elucidate the function of both *P. falciparum* and *P. vivax* StAR-related lipid transfer proteins, and importantly to determine whether antibodies to *P. vivax* PVX\_081550 are functionally protective or simply a useful marker of a child's overall immune status.

Our studies have confirmed that a large array of *P. vivax* merozoite antigens are targets of natural humoral immunity, and that antibodies to little-studied proteins may have equivalent or even stronger associations with reduced malaria risk in naturally exposed populations in comparison to current leading vaccine candidates. Further studies, including both in-depth evaluations of their association with protection in longitudinal cohort studies in other transmission settings and functional studies (to the extent this is currently possible for *P. vivax*), will be required to determine the potential of these proteins as vaccine candidates, markers of immune status, markers of cumulative exposure, or some combination thereof.

# **Supporting Information**

**S1 Text. Antibody kinetics.** (DOCX)

**S1 Fig. Comparison between antibody seroprevalence data and fits of the serocatalytic model.** Data are shown as point estimates (square) with 95% credible intervals (vertical bars) of seroprevalence in 5- or 10-year age bins. The model fit corresponding to the posterior median parameter estimates is shown with the solid line. (DOCX)

**S1** Table. Recombinant *P. vivax* proteins used in the study. (XLSX)

S2 Table. Associations between levels of IgG to 12 *P. vivax* merozoite proteins measured by ELISA and infection status, clinical symptoms, and socioeconomic indicators in Solomon Islanders.

(XLSX)

S3 Table. Associations between levels of IgG to 6 *P. vivax* merozoite proteins and age, exposure status, and infection status in Papua New Guinean children. (XLSX)

S4 Table. Correlation between IgG levels to 3 *P. vivax* merozoite antigens in Papua New Guinean children.

(XLSX)

S5 Table. Parameter estimates for serocatalytic models. Estimates are provided as median and 95% credible intervals of the posterior distribution. All parameters had uniform prior distributions:  $\lambda \sim U(0,10)$ ,  $\rho \sim U(0,1)$ . (XLSX)

S6 Table. Association between levels of IgG to *P. vivax* merozoite proteins and protection against clinical *P. falciparum* malaria episodes (all parasite densities) in Papua New Guinean children. (XLSX)

### Acknowledgments

We thank all patients and their families for participating in this study. We gratefully acknowledge Gavin J. Wright for his advice regarding expression of the *P. vivax* proteins used in this study and to the large field Solomon Island and Papua New Guinean field teams that assisted in the conduct of the fieldwork.

### **Author Contributions**

Conceived and designed the experiments: IM RMF JCR PS. Performed the experiments: CTF JBH. Analyzed the data: CTF MTW CSNLWS IM. Contributed reagents/materials/analysis tools: SS BK AWD AW CLK EL. Wrote the paper: CTF JBH CK JCR RMF IM.

#### References

- 1. WHO. World Malaria Report 2015. World Health Organization, Geneva, Switzerland. 2015.
- Alonso PL, Brown G, Arevalo-Herrera M, Binka F, Chitnis C, Collins F, et al. A research agenda to underpin malaria eradication. PLoS medicine. 2011; 8(1):e1000406. doi: <u>10.1371/journal.pmed.</u> <u>1000406</u> PMID: <u>21311579</u>; PubMed Central PMCID: PMC3026687.
- Feachem RG, Phillips AA, Hwang J, Cotter C, Wielgosz B, Greenwood BM, et al. Shrinking the malaria map: progress and prospects. Lancet. 2010; 376(9752):1566–78. doi: <u>10.1016/S0140-6736(10)61270-</u> <u>6</u> PMID: <u>21035842</u>; PubMed Central PMCID: PMC3044848.
- Betuela I, Rosanas-Urgell A, Kiniboro B, Stanisic DI, Samol L, de Lazzari E, et al. Relapses contribute significantly to the risk of Plasmodium vivax infection and disease in Papua New Guinean children 1–5 years of age. The Journal of infectious diseases. 2012; 206(11):1771–80. doi: <u>10.1093/infdis/jis580</u> PMID: <u>22966124</u>.
- Robinson LJ, Wampfler R, Betuela I, Karl S, White MT, Li Wai Suen CS, et al. Strategies for understanding and reducing the Plasmodium vivax and Plasmodium ovale hypnozoite reservoir in Papua New Guinean children: a randomised placebo-controlled trial and mathematical model. PLoS medicine. 2015; 12(10):e1001891. doi: <u>10.1371/journal.pmed.1001891</u> PMID: <u>26505753</u>; PubMed Central PMCID: PMCPMC4624431.
- Mueller I, Galinski MR, Baird JK, Carlton JM, Kochar DK, Alonso PL, et al. Key gaps in the knowledge of Plasmodium vivax, a neglected human malaria parasite. The Lancet Infectious diseases. 2009; 9 (9):555–66. PMID: <u>19695492</u>. doi: <u>10.1016/S1473-3099(09)70177-X</u>
- Bockarie MJ, Dagoro H. Are insecticide-treated bednets more protective against Plasmodium falciparum than Plasmodium vivax-infected mosquitoes? Malaria journal. 2006; 5:15. doi: <u>10.1186/1475-</u> 2875-5-15 PMID: 16504027; PubMed Central PMCID: PMC1388224.
- Luxemburger C, Perea WA, Delmas G, Pruja C, Pecoul B, Moren A. Permethrin-impregnated bed nets for the prevention of malaria in schoolchildren on the Thai-Burmese border. Transactions of the Royal Society of Tropical Medicine and Hygiene. 1994; 88(2):155–9. PMID: 8036656.
- 9. Sinka ME, Bangs MJ, Manguin S, Chareonviriyaphap T, Patil AP, Temperley WH, et al. The dominant Anopheles vectors of human malaria in the Asia-Pacific region: occurrence data, distribution maps and bionomic precis. Parasites & vectors. 2011; 4:89. doi: <u>10.1186/1756-3305-4-89</u> PMID: <u>21612587</u>; PubMed Central PMCID: PMC3127851.
- 10. Baird KJ, Maguire JD, Price RN. Diagnosis and treatment of Plasmodium vivax malaria. Advances in parasitology. 2012; 80:203–70. doi: 10.1016/B978-0-12-397900-1.00004–9 PMID: 23199489.
- Llanos-Cuentas A, Lacerda MV, Rueangweerayut R, Krudsood S, Gupta SK, Kochar SK, et al. Tafenoquine plus chloroquine for the treatment and relapse prevention of Plasmodium vivax malaria (DETEC-TIVE): a multicentre, double-blind, randomised, phase 2b dose-selection study. Lancet. 2014; 383 (9922):1049–58. doi: 10.1016/S0140-6736(13)62568-4 PMID: 24360369.
- mal ERACGoV. A research agenda for malaria eradication: vaccines. PLoS medicine. 2011; 8(1): e1000398. doi: <u>10.1371/journal.pmed.1000398</u> PMID: <u>21311586</u>; PubMed Central PMCID: PMC3026701.
- Fowkes FJ, Richards JS, Simpson JA, Beeson JG. The relationship between anti-merozoite antibodies and incidence of Plasmodium falciparum malaria: A systematic review and meta-analysis. PLoS medicine. 2010; 7(1):e1000218. doi: <u>10.1371/journal.pmed.1000218</u> PMID: <u>20098724</u>; PubMed Central PMCID: PMC2808214.
- Halbroth BR, Draper SJ. Recent developments in malaria vaccinology. Advances in parasitology. 2015; 88:1–49. doi: <u>10.1016/bs.apar.2015.03.001</u> PMID: <u>25911364</u>.

- Carlton JM, Adams JH, Silva JC, Bidwell SL, Lorenzi H, Caler E, et al. Comparative genomics of the neglected human malaria parasite Plasmodium vivax. Nature. 2008; 455(7214):757–63. doi: <u>10.1038/</u> <u>nature07327</u> PMID: <u>18843361</u>; PubMed Central PMCID: PMC2651158.
- Bozdech Z, Mok S, Hu G, Imwong M, Jaidee A, Russell B, et al. The transcriptome of Plasmodium vivax reveals divergence and diversity of transcriptional regulation in malaria parasites. Proceedings of the National Academy of Sciences of the United States of America. 2008; 105(42):16290–5. doi: <u>10.</u> <u>1073/pnas.0807404105</u> PMID: <u>18852452</u>; PubMed Central PMCID: PMC2571024.
- Chen JH, Jung JW, Wang Y, Ha KS, Lu F, Lim CS, et al. Immunoproteomics profiling of blood stage Plasmodium vivax infection by high-throughput screening assays. Journal of proteome research. 2010; 9(12):6479–89. doi: 10.1021/pr100705g PMID: 20949973.
- Lu F, Li J, Wang B, Cheng Y, Kong DH, Cui L, et al. Profiling the humoral immune responses to Plasmodium vivax infection and identification of candidate immunogenic rhoptry-associated membrane antigen (RAMA). J Proteomics. 2014; 102:66–82. doi: 10.1016/j.jprot.2014.02.029 PMID: 24607491.
- Finney OC, Danziger SA, Molina DM, Vignali M, Takagi A, Ji M, et al. Predicting antidisease immunity using proteome arrays and sera from children naturally exposed to malaria. Molecular & cellular proteomics: MCP. 2014; 13(10):2646–60. doi: <u>10.1074/mcp.M113.036632</u> PMID: <u>25023128</u>; PubMed Central PMCID: PMC4188993.
- Cutts JC, Powell R, Agius PA, Beeson JG, Simpson JA, Fowkes FJ. Immunological markers of Plasmodium vivax exposure and immunity: a systematic review and meta-analysis. BMC medicine. 2014; 12:150. doi: 10.1186/s12916-014-0150-1 PMID: 25199532; PubMed Central PMCID: PMC4172944.
- Mueller I, Shakri AR, Chitnis CE. Development of vaccines for Plasmodium vivax malaria. Vaccine. 2015; 33(52):7489–95. doi: 10.1016/j.vaccine.2015.09.060 PMID: 26428453.
- Hostetler JB, Sharma S, Bartholdson SJ, Wright GJ, Fairhurst RM, Rayner JC. A Library of Plasmodium vivax Recombinant Merozoite Proteins Reveals New Vaccine Candidates and Protein-Protein Interactions. PLoS Negl Trop Dis. 2015; 9(12):e0004264. doi: <u>10.1371/journal.pntd.0004264</u> PMID: <u>26701602</u>; PubMed Central PMCID: PMCPMC4689532.
- Crosnier C, Wanaguru M, McDade B, Osier FH, Marsh K, Rayner JC, et al. A library of functional recombinant cell-surface and secreted P. falciparum merozoite proteins. Molecular & cellular proteomics: MCP. 2013; 12(12):3976–86. doi: <u>10.1074/mcp.O113.028357</u> PMID: <u>24043421</u>; PubMed Central PMCID: PMCPMC3861738.
- Bushell KM, Sollner C, Schuster-Boeckler B, Bateman A, Wright GJ. Large-scale screening for novel low-affinity extracellular protein interactions. Genome Res. 2008; 18(4):622–30. doi: <u>10.1101/gr.</u> <u>7187808</u> PMID: <u>18296487</u>; PubMed Central PMCID: PMCPMC2279249.
- Kellar KL, Kalwar RR, Dubois KA, Crouse D, Chafin WD, Kane BE. Multiplexed fluorescent beadbased immunoassays for quantitation of human cytokines in serum and culture supernatants. Cytometry. 2001; 45(1):27–36. PMID: <u>11598944</u>.
- 26. Waltmann A, Darcy AW, Harris I, Koepfli C, Lodo J, Vahi V, et al. High Rates of Asymptomatic, Submicroscopic Plasmodium vivax Infection and Disappearing Plasmodium falciparum Malaria in an Area of Low Transmission in Solomon Islands. PLoS neglected tropical diseases. 2015; 9(5):e0003758. doi: 10.1371/journal.pntd.0003758 PMID: 25996619; PubMed Central PMCID: PMC4440702.
- Lin E, Kiniboro B, Gray L, Dobbie S, Robinson L, Laumaea A, et al. Differential patterns of infection and disease with P. falciparum and P. vivax in young Papua New Guinean children. PloS one. 2010; 5(2): e9047. doi: 10.1371/journal.pone.0009047 PMID: 20140220; PubMed Central PMCID: PMC2816213.
- Mueller I, Schoepflin S, Smith TA, Benton KL, Bretscher MT, Lin E, et al. Force of infection is key to understanding the epidemiology of Plasmodium falciparum malaria in Papua New Guinean children. Proc Natl Acad Sci U S A. 2012; 109(25):10030–5. doi: <u>10.1073/pnas.1200841109</u> PMID: <u>22665809</u>; PubMed Central PMCID: PMCPMC3382533.
- Koepfli C, Colborn KL, Kiniboro B, Lin E, Speed TP, Siba PM, et al. A high force of plasmodium vivax blood-stage infection drives the rapid acquisition of immunity in papua new guinean children. PLoS neglected tropical diseases. 2013; 7(9):e2403. doi: <u>10.1371/journal.pntd.0002403</u> PMID: <u>24040428</u>; PubMed Central PMCID: PMC3764149.
- Piriou E, Kimmel R, Chelimo K, Middeldorp JM, Odada PS, Ploutz-Snyder R, et al. Serological evidence for long-term Epstein-Barr virus reactivation in children living in a holoendemic malaria region of Kenya. J Med Virol. 2009; 81(6):1088–93. doi: <u>10.1002/jmv.21485</u> PMID: <u>19382256</u>; PubMed Central PMCID: PMC3134942.
- Corran P, Coleman P, Riley E, Drakeley C. Serology: a robust indicator of malaria transmission intensity? Trends in parasitology. 2007; 23(12):575–82. doi: <u>10.1016/j.pt.2007.08.023</u> PMID: <u>17988945</u>.
- Yman V, White MT, Rono J, Arca B, Osier FH, Troye-Blomberg M, et al. Antibody acquisition models: A new tool for serological surveillance of malaria transmission intensity. Sci Rep. 2016; 6:19472. doi: <u>10.</u> <u>1038/srep19472</u> PMID: <u>26846726</u>.

- Giraldo J, Vivas NM, Vila E, Badia A. Assessing the (a)symmetry of concentration-effect curves: empirical versus mechanistic models. Pharmacol Ther. 2002; 95(1):21–45. PMID: 12163126.
- 34. Stanisic DI, Javati S, Kiniboro B, Lin E, Jiang J, Singh B, et al. Naturally acquired immune responses to P. vivax merozoite surface protein 3alpha and merozoite surface protein 9 are associated with reduced risk of P. vivax malaria in young Papua New Guinean children. PLoS neglected tropical diseases. 2013; 7(11):e2498. doi: <u>10.1371/journal.pntd.0002498</u> PMID: <u>24244763</u>; PubMed Central PMCID: PMC3828159.
- Mueller I, Galinski MR, Tsuboi T, Arevalo-Herrera M, Collins WE, King CL. Natural acquisition of immunity to Plasmodium vivax: epidemiological observations and potential targets. Advances in parasitology. 2013; 81:77–131. doi: <u>10.1016/B978-0-12-407826-0.00003–5</u> PMID: <u>23384622</u>.
- Elliott SR, Fowkes FJ, Richards JS, Reiling L, Drew DR, Beeson JG. Research priorities for the development and implementation of serological tools for malaria surveillance. F1000Prime Rep. 2014; 6:100. doi: 10.12703/P6-100 PMID: 25580254; PubMed Central PMCID: PMCPMC4229730.
- Longley RJ, Sattabongkot J, Mueller I. Insights into the naturally acquired immune response to Plasmodium vivax malaria. Parasitology. 2016; 143(2):154–70. doi: <u>10.1017/S0031182015000670</u> PMID: <u>26864134</u>.
- Osier FH, Mackinnon MJ, Crosnier C, Fegan G, Kamuyu G, Wanaguru M, et al. New antigens for a multicomponent blood-stage malaria vaccine. Sci Transl Med. 2014; 6(247):247ra102. doi: <u>10.1126/</u> <u>scitranslmed.3008705</u> PMID: <u>25080477</u>.
- 39. Richards JS, Arumugam TU, Reiling L, Healer J, Hodder AN, Fowkes FJ, et al. Identification and prioritization of merozoite antigens as targets of protective human immunity to Plasmodium falciparum malaria for vaccine and biomarker development. Journal of immunology. 2013; 191(2):795–809. doi: 10.4049/jimmunol.1300778 PMID: 23776179; PubMed Central PMCID: PMC3702023.
- 40. Li J, Ito D, Chen JH, Lu F, Cheng Y, Wang B, et al. Pv12, a 6-Cys antigen of Plasmodium vivax, is localized to the merozoite rhoptry. Parasitol Int. 2012; 61(3):443–9. doi: <u>10.1016/j.parint.2012.02.008</u> PMID: <u>22394409</u>.
- Cheng Y, Lu F, Tsuboi T, Han ET. Characterization of a novel merozoite surface protein of Plasmodium vivax, Pv41. Acta tropica. 2013; 126(3):222–8. doi: <u>10.1016/j.actatropica.2013.03.002</u> PMID: <u>23499861</u>.
- Tonkin ML, Arredondo SA, Loveless BC, Serpa JJ, Makepeace KA, Sundar N, et al. Structural and biochemical characterization of Plasmodium falciparum 12 (Pf12) reveals a unique interdomain organization and the potential for an antiparallel arrangement with Pf41. J Biol Chem. 2013; 288(18):12805–17. doi: 10.1074/jbc.M113.455667 PMID: 23511632; PubMed Central PMCID: PMCPMC3642325.
- van Ooij C, Withers-Martinez C, Ringel A, Cockcroft S, Haldar K, Blackman MJ. Identification of a Plasmodium falciparum phospholipid transfer protein. J Biol Chem. 2013; 288(44):31971–83. doi: <u>10.1074/jbc.M113.474189</u> PMID: <u>24043620</u>; PubMed Central PMCID: PMCPMC3814793.
- Fan YT, Wang Y, Ju C, Zhang T, Xu B, Hu W, et al. Systematic analysis of natural antibody responses to P. falciparum merozoite antigens by protein arrays. Journal of proteomics. 2013; 78:148–58. doi: <u>10.</u> <u>1016/j.jprot.2012.11.020</u> PMID: <u>23201113</u>.
- 45. Aurrecoechea C, Brestelli J, Brunk BP, Dommer J, Fischer S, Gajria B, et al. PlasmoDB: a functional genomic database for malaria parasites. Nucleic Acids Res. 2009; 37:D539–43. doi: <u>10.1093/nar/gkn814</u> PMID: <u>18957442</u>